Author's response to reviews

Title: Assessing fetal growth impairments based on family data as a tool for identifying high-risk babies. An example with neonatal mortality

Authors:

Carsten B Pedersen (cbp@ncrr.dk)
Yuelian Sun (YS@SOCI.AU.DK)
Mogens Vestergaard (MOGENS.VESTERGAARD@ALM.AU.DK)
Jorn Olsen (jo@ucla.edu)
Olga Basso (bassoo2@niehs.nih.gov)

Version: 5 Date: 24 November 2007

Author's response to reviews:

Dear Editor

Many thanks for your comments concerning the formatting requirements.

We have changed the paper according to all requested changes. However, you state that the references jump from 33 to 39. I have double checked all references in the paper, but note that our paper includes only 33 references.